Author's response to reviews

Title: Dietary intake of fish, omega-3, omega-6 polyunsaturated fatty acids and vitamin D and the prevalence of psychotic-like symptoms in a cohort of 33 000 women from the general population.

Authors:

Maria Hedelin (maria.hedelin@ki.se)
Marie Löf (marie.lof@ki.se)
Marita Olsson (marita.olsson@ki.se)
Tommy Lewander (Tommy.Lewander@neuro.uu.se)
Björn Nilsson (Bjorn.Nilsson@neuro.uu.se)
Christina Hultman (Christina.Hultman@ki.se)
Elisabete Weiderpass (eliwei@ki.se)

Version: 3 Date: 12 May 2010

Author's response to reviews:

Dear Editor,

We have now provided both the major and minor revisions requested. Please let us know if the formatting changes we made are acceptable.

Thank you.